Author's response to reviews

Title: Chromosomal alterations in exfoliated urothelial cells from bladder cancer cases and healthy men: a prospective screening study

Authors:

Nadine Bonberg (bonberg@ipa-dguv.de)
Beate Pesch (pesch@ipa-dguv.de)
Thomas Behrens (behrens@ipa-dguv.de)
Georg Johnen (johnen@ipa-dguv.de)
Dirk Taeger (taeger@ipa-dguv.de)
Katarzyna Gawrych (gawrych@ipa-dguv.de)
Christian Schwentner (christian.schwentner@med.uni-tuebingen.de)
Harald Wellhäusser (harald.wellhaeusser@bgrci.de)
Matthias Kluckert (matthias.kluckert@bgrci.de)
Gabriele Leng (gabriele.leng@currenta.de)
Michael Nasterlack (nasterlack@arcor.de)
Christoph Oberlinner (christoph.oberlinner@basf.com)
Arnulf Stenzl (arnulf.stenzl@med.uni-tuebingen.de)
Thomas Bruening (bruening@ipa-dguv.de)

Version: 4
Date: 31 October 2014

Author's response to reviews:

Dear Editor,

Please, find our copy-edited manuscript submitted as clean version. We have asked a native speaker to improve the language and marked the changes in the manuscript submitted as additional material. We hope that the revised manuscript will be suitable for publication in your Journal.

Sincerely yours,

Beate Pesch